| SPS | | | | | | | |
|--|--|---------|------------------------|-------------|----------|----------|----|
| Machine C | oordinator la | st week | | | | | |
| Machine Coordinator this week | | | Johan, James, Stephane | | | | |
| Beam Scheduled | | | | | | | |
| LHC | No | NA | No | AWAKE | No | HiRadMat | No |
| Beam Availability by Destination (AFT) | | | | | | | |
| LHC | % | NA | % | AWAKE | % | HiRadMat | % |
| Facility Status | | | | | | | |
| Summary | Patrol done. Test HV on the main circuit done without issue Test of the new SMD13 configuration ongoing, probably a problem on the masse which created a lot of noise under investigation and maybe also an electronic card problem. Parasitic HWC tests ongoing IST finished with success No showstopper for the moment | | | | | | |
| Issues | Link between BA80 BLM crate and BA2 BIC not connected under investigation MSE on West extraction solved On SMD13 noise problem under investigation | | | | | | |
| Plans | Start the HWC for 3 weeks | | | | | | |
| Intervention Request | | | | | | | |
| Yes / No | Duration | | | Preferred d | ate/time | | |
| Reason | | | | | | | |
| Impact | | | | | | | |